Author's response to reviews

Title: Immigrant women's experiences of maternity care services in Canada: a protocol for systematic review using a narrative synthesis.

Authors:

Gina MA Higginbottom (gina.higginbottom@ualberta.ca)
Morgan Myfanwy (morgan.myfanwy@kcl.ac.uk)
Jayantha Dassanayake (jayantha.dassanayake@monash.edu)
Helgi Eyford (heyford@shaw.ca)
Mirande Alexandre (mirande.alexandre@cic.gc.ca)
Yvonne Chiu (yvonne.chiu@shaw.ca)
Joan Forgeron (joan.forgeron@albertahealthservices.ca)
Deb Kocay (debbie.kocay@phac-aspc.gc.ca)

Version: 2 Date: 30 April 2012

Author's response to reviews:

Dear Editor,

Thank you very much for agreeing to publish our protocol pending a couple very minor changes. We have uploaded the manuscript again with the two changes requested, both highlighted in red type.

Most sincerely,
Gina Higginbottom